Operability Report 

2/7/2007
Accelerator
Energy shifts after FSD trips
· Run tune beam, adjust energy lock cavities to recover orbits
· BCM faults, AT07 BLM trips
· Caused by a change in the beam optics after a LEM (low energy effect)
· Short Term Fix – Cycle NL quads after running LEM, Use scale skew knob to fix spot on 1A SLM
· Also check spots on AT06 and Lambertson viewers
· Long Term Fix – Update LEM software
· Tight aperture in AT line
· Try steering with first few correctors in AT line (H & V) while looking at beam loss (BCM and BSM) and halo rates
· Arc 4 box supply current instabilities causing BCM or BLM faults and high halo
· Effects MAX4S02 and possibly MAX4T05
· Power factor correction capacitor (PFCC) reinstalled by EES
· Calibrated the transistor bank voltage read back ( to help diagnose)

· Checked all aux. PS voltages (nominal)

· Checked -15 Volt read back fluctuations. PS is solid but read back is jumping around. This is not causing a problem.

· Changed the SCR firing board
· Tightened the feedback connections for the transistor bank voltage control loop
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· 1 pass 362 MeV

· Up to 40 uA

· Weekly Mott measurements

· Halo rates
· Position asymmetry distribution

· FFB position and energy





· Low energy optics so beam is Very Sensitive
· High halo counts
· Dispersion leaking out of hall C arc
· Fixed with 3C12 quad
· For dispersion leaking into hall C, try the quad at 1R10
· Also can be caused by scraping in the AT line due to arc 4 box supply
· Fixed by steering at AT02, AT03, AT04
· Also check aperture using the viewer at AT06
· Position asymmetry growth at end of hall C line
· Caused by bad aspect ratio at G0 target
· Fixed using quad at 3C16, 3C17, or 3C19 and online asymmetry tool  
